AER336S Scientific Computing: Introduction and Course Outline

January 5, 2001 Professor D.W. Zingg (667-7709, dwz@oddjob.utias.utoronto.ca, http://goldfinger.utias.utoronto.ca/~dwz/) Teaching Assistant: Todd Chisholm (667-7887, todd@oddjob.utias.utoronto.ca)

Textbook: C.F. Van Loan, Introduction to Scientific Computing, second edition, Prentice-Hall, 2000. Reference Book: G.H. Golub and J.M. Ortega, Scientific Computing and Differential Equations, Academic Press, 1992. Mark Breakdown: Assignments 25%, Tests: 25%, Final exam 50%. Tests will be held on Feb. 12 and March 26. The final exam will be type X, i.e., "open-book," calculator type 1.

Course Outline

1. Interpolation 2. Integration 3. Linear Systems 4. Least Squares Fitting 5. Nonlinear Equations and Optimization 6. The Initial Value Problem 7. Partial Differential Equations 8. Relaxation Methods